Author's response to reviews

**Title:** Rheumatoid Arthritis patients are not in a position to plan their near future. Intra-subject correlation in personal assessment of health levels over two month - Results of a survey

**Authors:**

- Rudolf Puchner (rudolf.puchner@cc-net.at)
- Hans-Peter Brezinschek (hans-peter.brezinsek@meduni-graz.at)
- Josef Fritz (josef.fritz@i-med.ac.at)
- Manfred Herold (manfred.herold@i-med.ac.at)
- Monika Mustak (monika.mustak@wienkav.at)
- Thomas Nothnagl (thomas.nothnagl@stockerau.lknoe.at)
- Stephan E Puchner (stephan.puchner@meduniwien.ac.at)
- Andrea Studnick-Benke (a.studnicka-benke@salk.at)
- Burkhard F Leeb (leeb.rheuma@aon.at)

**Version:** 2

**Date:** 14 April 2014

**Author's response to reviews:**

The mail addresses of all authors have been included;
The Abstract was modified as proposed.
The page numbers and line numbers have been included.
Figure legends have been included and the figures removed from the main document.
The two questionnaires utilized will be uploaded with all the other revised material.